Author's response to reviews

Title: The personal and national costs of mental health conditions: impacts on income, taxes, government support payments due to lost labour force participation

Authors:

Deborah J Schofield (deborah.schofield@ctc.usyd.edu.au)
Rupendra N Shrestha (rupendra.shrestha@ctc.usyd.edu.au)
Richard Percival (richard.percival@natsem.canberra.edu.au)
Megan E Passey (megan.passey@ncahs.health.nsw.gov.au)
Emily J Callander (emily.callander@ctc.usyd.edu.au)
Simon J Kelly (simon@kellyresearch.com.au)

Version: 3 Date: 23 March 2011

Author's response to reviews:

The following formatting changes have been made:

----------------------------------

Major revisions (we require the author to make these changes)

Footnotes: We are unable to accommodate footnotes in the full text version of the manuscript. We suggest that you incorporate the footnotes into the text of the manuscript. Alternatively, you may wish to make an appendix section containing your footnotes.

Conclusions - Please include a Conclusions section after the Discussion.

Tables: you have cited 4 tables in the manuscript yet only 3 have been included, please either include this missing table in the manuscript file or correct this citation

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Affiliations: Please remember to include the country in the affiliation details

Declaration of interests: please rename this competing interests and move to above the authors contributions

Introduction: please rename this Background

Highlighting: please remove all highlighting from the manuscript